Author's response to reviews

Title: A protocol for an individualised, facilitated and sustainable approach to implementing current evidence in preventing falls in residential aged care facilities

Authors:

Betty Haralambous (b.haralambous@nari.unimelb.edu.au)
Terry P Haines (Terrence.Haines@med.monash.edu.au)
Keith Hill (k.hill@nari.unimelb.edu.au)
Kirsten Moore (k.moore@nari.unimelb.edu.au)
Jennifer Nitz (j.nitz@uq.edu.au)
Andrew Robinson (Andrew.Robinson@utas.edu.au)

Version: 4 Date: 12 February 2010

Author's response to reviews:

Major revisions have been addressed as requested

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution